Author's response to reviews

Title: Knowledge of vitamin D and perceptions and attitudes towards sunlight among Chinese middle-aged and elderly women: a population survey in Hong Kong

Authors:

Annie WC Kung (awckung@hkucc.hku.hk)
Kar-Kui Lee (kklee@graduate.hku.hk)

Version: 7 Date: 5 September 2006

Author's response to reviews:

As instructed, 'Consent was signed by individuals willing to participate in the study. The study was approved by the Ethics Committee of the University of Hong Kong and was conducted according to the Declaration of Helsinki.' was added at the end of the 'Method' section.